Author's response to reviews

Title: Exercise intervention to prevent falls and enhance mobility in community dwellers after stroke: a protocol for a randomised controlled trial

Authors:

Catherine M Dean (C.Dean@usyd.edu.au)
Chris Rissel (criss@email.cs.nsw.gov.au)
Michelle Sharkey (info@strokensw.org.au)
Catherine Sherrington (csherrington@george.org.au)
Robert G Cumming (bobc@health.usyd.edu.au)
Ruth N Barker (ruth.barker@jcu.edu.au)
Stephen R Lord (S.lord@powmri.edu.au)
Sandra D O'Rourke (s.orourke@usyd.edu.au)
Catherine Kirkham (c.kirkham@usyd.edu.au)

Version: 8 Date: 22 July 2009

Author's response to reviews:

Dear editor

manuscript has been proof read, slight changes were made to the headings throughout the manuscript.

thanks

Cath Dean